Author's response to reviews

Title: Early menopause: Association with tobacco smoking, coffee consumption and other lifestyle factors - a cross-sectional study

Authors:

Thea F Mikkelsen (theam@studmed.uio.no)
Sidsel Graff-Iversen (sgri@fhi.no)
Johanne Sundby (johanne.sundby@medisin.no)
Espen Bjertness (espen.bjertness@medisin.no)

Version: 3 Date: 20 June 2007

Author's response to reviews: see over
Dear Editor of BMC Public Health

The authors thank you again for your interest in our manuscript entitled “Early Menopause: Association with tobacco smoking, coffee consumption and other lifestyle factors—a cross-sectional study”. Please find enclosed our revised manuscript: We have revised the manuscript according to your recommendations. The paragraph about the link between education, genetics and intrauterine life is now deleted and we have added information about the ethical approval in the methods section.

Thank you for your attention!
On behalf of all co-authors who have read and approved the changes

Thea Falkenberg Mikkelsen